## **PLOS PATHOGENS**



### GOPEN ACCESS

**Citation:** Krücken J, Holden-Dye L, Keiser J, Prichard RK, Townson S, Makepeace BL, et al. (2021) Development of emodepside as a possible adulticidal treatment for human onchocerciasis— The fruit of a successful industrial–academic collaboration. PLoS Pathog 17(7): e1009682. https://doi.org/10.1371/journal.ppat.1009682

**Editor:** James B. Lok, University of Pennsylvania, UNITED STATES

Published: July 22, 2021

**Copyright:** © 2021 Krücken et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: Bayer AG funded editorial support (Highfield Communication, Oxford, United Kingdom). The Bill & Melinda Gates Foundation covers the publication fees. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** I have read the journal's policy and want to declare the following conflicts of interest. The authors DK, SH, AH were employees of Bayer Animal Health GmbH, Germany. Bayer Animal Health GmbH was a company that developed and commercialized veterinary

REVIEW

## Development of emodepside as a possible adulticidal treatment for human onchocerciasis—The fruit of a successful industrial–academic collaboration

Jürgen Krücken<sup>1</sup>, Lindy Holden-Dye<sup>2</sup>, Jennifer Keiser<sup>3,4</sup>, Roger K. Prichard<sup>5</sup>, Simon Townson<sup>6</sup>, Benjamin L. Makepeace<sup>7</sup>, Marc P. Hübner<sup>8,9</sup>, Steffen R. Hahnel<sup>10</sup>, Ivan Scandale<sup>11</sup>, Achim Harder<sup>12</sup>, Daniel Kulke<sup>13</sup>\*

1 Institute for Parasitology and Tropical Veterinary Medicine, Freie Universität Berlin, Berlin, Germany, 2 School of Biological Sciences, Institute for Life Sciences, University of Southampton, Southampton, United Kingdom, 3 Department of Medical Parasitology and Infection Biology, Swiss Tropical and Public Health Institute, Basel, Switzerland, 4 University of Basel, Basel, Switzerland, 5 Institute of Parasitology, McGill University, Sainte Anne-de-Bellevue, Quebec, Canada, 6 The Griffin Institute, Northwick Park and St. Mark's Hospital, Harrow, United Kingdom, 7 Institute of Infection, Veterinary & Ecological Sciences, University of Liverpool, Liverpool, United Kingdom, 8 Institute for Medical Microbiology, Immunology and Parasitology, University Hospital Bonn, Bonn, Germany, 9 German Center for Infection Research (DZIF), partner site Bonn-Cologne, Bonn, Germany, 10 Elanco Animal Health, Research & Exploratory Development, Monheim, Germany, 11 Drugs for Neglected Disease initiative, Geneva, Switzerland, 12 Independent Researcher, Cologne, Germany, 13 Department of Biomedical Sciences, Iowa State University, Ames, Iowa, United States of America

\* dkulke@iastate.edu

### Abstract

Current mass drug administration (MDA) programs for the treatment of human river blindness (onchocerciasis) caused by the filarial worm Onchocerca volvulus rely on ivermectin, an anthelmintic originally developed for animal health. These treatments are primarily directed against migrating microfilariae and also suppress fecundity for several months, but fail to eliminate adult O. volvulus. Therefore, elimination programs need time frames of decades, well exceeding the life span of adult worms. The situation is worsened by decreased ivermectin efficacy after long-term therapy. To improve treatment options against onchocerciasis, a drug development candidate should ideally kill or irreversibly sterilize adult worms. Emodepside is a broad-spectrum anthelmintic used for the treatment of parasitic nematodes in cats and dogs (Profender and Procox). Our current knowledge of the pharmacology of emodepside is the result of more than 2 decades of intensive collaborative research between academia and the pharmaceutical industry. Emodepside has a novel mode of action with a broad spectrum of activity, including against extraintestinal nematode stages such as migrating larvae or macrofilariae. Therefore, emodepside is considered to be among the most promising candidates for evaluation as an adulticide treatment against onchocerciasis. Consequently, in 2014, Bayer and the Drugs for Neglected Diseases initiative (DNDi) started a collaboration to develop emodepside for the treatment of patients suffering from the disease. Macrofilaricidal activity has been demonstrated in various models, including Onchocerca ochengi in cattle, the parasite most closely related to O. volvulus.

medicines including anthelmintics. The remaining authors of this manuscript have declared no competing interests.

Emodepside has now successfully passed Phase I clinical trials, and a Phase II study is planned. This Bayer–DNDi partnership is an outstanding example of "One World Health," in which experience gained in veterinary science and drug development is translated to human health and leads to improved tools to combat neglected tropical diseases (NTDs) and shorten development pathways and timelines in an otherwise neglected area.

#### Author summary

*Onchocerca volvulus* is the causative agent of human river blindness, and current elimination programs rely on the use of ivermectin to kill microfilariae. Since no adulticidal drug is available and adult worms have a life span of up to 15 years, elimination programs need to be sustained over several decades. Emodepside is an anthelmintic that is licensed as a dewormer for cats and dogs. Due to its ability to eliminate nematodes located in various extraintestinal host tissues, including migrating larvae and adult filarial worms, it is considered to be an excellent candidate for the treatment of onchocerciasis. Intense collaboration between academia and the pharmaceutical industry has led to a deep understanding of the novel mode of action of the drug and of its parasite target spectrum. Phase I clinical trials with emodepside have demonstrated its safety and adulticide activity against the closely related cattle parasite *Onchocerca ochengi*. Currently, Phase II clinical trials are planned to confirm that emodepside, developed initially to improve animal health, has also the potential to improve human health by tackling a very important neglected tropical disease (NTD).

# Elimination of onchocerciasis—Current status and the need for novel treatment strategies

Onchocerciasis, also known as river blindness, is a parasitic, vector-borne disease caused by the filarial nematode *Onchocerca volvulus*. Filarial diseases such as onchocerciasis and lymphatic filariasis place a tremendous burden on society, with more than 1 billion of the world's poorest populations at risk of infection [1,2]. Despite *O. volvulus* being one of the most prevalent species (see <u>S1 Fig</u> for global distribution), with 21 million people infected and approximately 90 million people at risk of infection in sub-Saharan Africa, plus 14.6 million people suffering from skin disease and approximately 0.77 million people debilitated by severe vision impairment or blindness, its elimination is considered to be possible [3–6].

Four life stages of *O. volvulus* live in humans. Infective third-stage larvae (L3) transmitted by the vector (i.e., a species of the genus *Simulium* [black flies]) undergo 2 molts to develop via fourth-stage larvae (L4) into juvenile adults. These mature to become reproductively competent adults (male and female macrofilariae) within around 1 year and have an estimated mean reproductive life span of up to 15 years [7]. Macrofilariae reside primarily in subcutaneous and deep tissue nodules and produce progeny (microfilariae) that live for approximately 1 year, predominantly in the subepidermal layer of the dermis. During the blood meal, female *Simulium* take up microfilariae, which then develop via second-stage larvae (L2) into infective L3 in the insect's thoracic muscles to complete the life cycle of the parasite [8] (see S2 Fig for illustration of the life cycle).

Onchocerciasis causes severe itching, disfiguring skin lesions, and depigmentation, along with musculoskeletal pain, reduced body mass index, and decreased work productivity [9,10]. The most severe complications attributed to onchocerciasis are severe visual impairment in up

to 500,000 cases, blindness in about 270,000 cases, and reduced life expectancy [6,11], although these figures are considered to underestimate the true magnitude of the socioeconomic burden of onchocerciasis [9]. The disease is the second most frequent cause of infectious blindness after trachoma, with almost all cases seen in sub-Saharan Africa [3].

Mass drug administration (MDA) is a means of delivering safe and inexpensive essential medicines based on the principles of preventive chemotherapy, in which populations or sub-populations are offered treatment without individual diagnoses. MDA in endemic areas aims to prevent and alleviate symptoms and morbidity on the one hand and reduce transmission on the other, together improving global health. At present, MDA with ivermectin (Mectizan, Merck, Kenilworth, New Jersey, USA) is the current measure to achieve elimination of onchocerciasis [12–14]. Since ivermectin does not kill the adult worms, it has to be given once or twice per year and is contraindicated in individuals harboring high numbers of *Loa loa* microfilariae due to the risk of life-threatening adverse events [15–17].

Insufficient coverage is one of the main factors hindering progress toward elimination. Reasons for this are many, including difficult to treat areas (*L. loa* coinfection, ongoing conflicts), lack of financial resources, and inadequate political engagement. Additionally, program fatigue in MDA has been reported after years of implementation, and an individual in an endemic area may find repeated MDA inconvenient or lose confidence in the MDA campaign. Furthermore, alarming reports of reduced drug efficacy in Ghana and Cameroon may also signal the development of resistance [18,19]. This situation, coupled with the risk of ivermectin resistance, presents an urgent need for a novel anthelmintic drug that preferably possesses macrofilaricidal or permanent sterilizing activity.

Drug development is handicapped by high attrition rates, and many promising molecules fail during preclinical development or in subsequent toxicological, safety, and efficacy testing; thus, research and development (R&D) costs in aggregate are very high. The level of investment into R&D for new products for neglected tropical diseases (NTDs), as reported in the annual Global Funding of Innovation for Neglected Diseases (G-FINDER) surveys, suggests that few NTD areas receive anywhere near the level of funding required; moreover, that funding, when it is available, is rarely allocated in a manner likely to move products through the pipeline to patients [20]. Therefore, no dedicated drug development pipeline for human filariasis is in place, and it is thus essential that stakeholders, funders, industry, academics, and nongovernmental organizations adopt a cooperative approach, sharing responsibility to reduce risks and overcome existing obstacles. Joint efforts should also be made to cut the cost of R&D for new drugs for NTDs.

The semisynthetic anthelminthic emodepside (synonyms: PF1022-221, BAY 44–4400, or bismorpholino–cyclooctadepsipeptide) and its parent fermentation product PF1022A are members of the N-methylated cyclooctadepsipeptides (Fig\_1). Emodepside is characterized by 2 morpholine rings in para-position of each of the 2 (R)-phenyllactic acids (Fig\_1A), which increase solubility and improve bioavailability in comparison to its natural precursor PF1022A (Fig\_1B) [21]. This precursor is obtained from a fungus (*Rosellinia* sp. PF1022), which is part of the microflora of the leaves of *Camellia japonica* [22]. Emodepside is registered as a combination product with praziquantel (Profender) for the treatment of cats and dogs infected with, or at risk of, infection with nematodes and cestodes and as a combination product with toltrazuril (Procox) for the treatment of puppies with demonstrated or suspected infection with nematodes and coccidia.

Emodepside has been shown to be microfilaricidal, as well as macrofilaricidal, and has a unique mechanism of action relative to all other anthelmintic drugs [22,23]. The advantage of emodepside over ivermectin is that it targets multiple life cycle stages of filariae, including the adult worms [24]. In the case of the genus *Onchocerca*, the in vitro motility of macrofilariae is



Fig 1. Chemical structures of the semisynthetic emodepside (A) and its precursor PF1022A (B) purified from *Rosellinia* spp. PF1022.

https://doi.org/10.1371/iournal.ppat.1009682.g001

affected at lower concentrations than that of the microfilariae [24]. Targeting macrofilariae is expected to result in a reduction in the number of MDA treatment cycles required to break transmission and cure patients of both migrating microfilariae as well as macrofilariae.

Since standard treatment with ivermectin is contraindicated in patients coinfected with *L*. *loa*, availability of an anthelmintic that can be used in regions where loiasis and onchocerciasis are co-endemic would add considerable value toward the goal of eliminating onchocerciasis [15-17]. As to date, emodepside has not been given to patients infected with *O. volvulus*, there is obviously no data on patients being coinfected with *L. loa*. Thus, after determining the emodepside efficacy and safety profile in patients infected with *O. volvulus*, a next step might be to test it in patients coinfected with *L. loa* (low microfilariae level) to evaluate the potential benefit in those patients.

The in vivo anthelmintic efficacy of the fermentation product PF1022A was originally demonstrated against the gastrointestinal nematode *Ascaridia galli* in chickens in 1992 [25], followed by reports of anthelminthic effects against a wide range of gastrointestinal nematodes in rats [26–28], mice [29–31], jirds [32], dogs, horses, sheep, and cattle [27]. In addition, emodepside showed efficacy against further species of parasitic nematodes in a variety of hosts, including cerebral and other extraintestinal life cycle stages [22,33–41]. Emodepside has an excellent safety profile, with neurotoxicological side effects only reported rarely in dogs with the multidrug resistance 1 (*mdr-1*) mutation [42]. More importantly, it has been shown to have anthelmintic resistance–breaking properties against several nematode isolates resistant to closantel, fenbantel, fenbendazole, levamisole, and ivermectin in sheep and cattle and against a multidrug-resistant *Ancylostoma caninum* isolate in dogs [43,44].

In addition to its well-known use as an anthelmintic in cats and dogs, the microfilaricidal and macrofilaricidal effects of emodepside have led to the compound being considered as a promising drug development candidate for the treatment of human helminth diseases, including onchocerciasis [22,45,46]. As a result, emodepside is currently being evaluated for the

treatment of human onchocerciasis within the scope of a drug development partnership between the Drugs for Neglected Diseases initiative (DNDi) and Bayer [47]. Phase I clinical trials have already been completed successfully [48], and a Phase II trial is planned in 2021.

Here, we outline the development of emodepside for human onchocerciasis, review the elucidation of its novel mode of action and its efficacy against filarial nematodes, both in vitro and in vivo, and describe the initial studies with this encouraging agent in humans.

### Emodepside—Novel mode of action and broad-spectrum activity

#### History of the cyclooctadepsipeptide anthelmintics

A flow diagram presenting the most important steps leading from the isolation of PF1022A to human trials evaluating safety and efficacy of the drug is available in Fig 2. In 1992, researchers from Meiji Seika Kaisha (Tokyo, Japan) isolated a new family of N-methylated cyclooctadepsipeptides [49]. The fungus *Rosellinia* spp. PF1022 produces 8 different cyclooctadepsipeptides, designated PF1022A, B, C, D, E, F, G, and H, although PF1022A is produced in the highest amounts by far [50]. PF1022A is also the most potent anthelmintic among the PF1022 members; its anthelmintic activities have been summarized in a previous review [22].

Since the efficacy of the parent compound PF1022A is largely limited to intestinal nematode stages in different host species, further research was focused on identifying cyclooctadepsipeptides with efficacy against extraintestinal stages (including blood and tissue stages). For this purpose, a screening program involving Bayer Animal Health (Leverkusen, Germany), Bayer Crop Science (Monheim, Germany), Meiji Seika Kaisha, and Fujisawa Pharmaceutical (Japan) was initiated. A patent application was subsequently filed by Fujisawa Pharmaceutical for a bis-morpholino derivative of PF1022A, which was later named emodepside [22].

#### Elucidation of the mode of action

#### Ionotropic $\gamma$ -aminobutyric acid receptors

Initial studies investigating the mechanism of action of PF1022A found high binding affinity of the cyclooctadepsipeptide to *Ascaris suum* cell membranes, suggesting that there are specific PF1022A binding sites in the membrane preparations that are responsible for its nematocidal activity [51]. PF1022A was also shown to displace radioactive γ-aminobutyric acid (GABA) from *A. suum* muscle cells, suggesting an interaction of PF1022A with ionotropic GABA<sub>A</sub> receptors [52]. However, PF1022A does not appear to act like a GABA agonist in *A. suum* muscle cells since it does not produce an increase in muscle membrane conductance similar to GABA [53,54]. Nevertheless, a loss-of-function mutation of the only GABA<sub>A</sub> receptor gene *unc-49* in the *Caenorhabditis elegans* genome has been shown to cause decreased susceptibility to emodepside [55].

#### Latrophilin receptors

The first target of cyclooctadepsipeptides identified in nematodes was the *Haemonchus contor*tus latrophilin-like receptor (LAT-1) [56]. An affinity screen of an *H. contortus* cDNA expression library identified a G protein–coupled receptor (GPCR) that bound PF1022A [56]. Expression of the full-length *H. contortus* LAT-1 in HEK293 cells further supported its potential to act as a receptor for latrotoxin since it initiated downstream Ca<sup>2+</sup> influx through Cd<sup>2+</sup> and nifedipine blockable channels (L-type Ca<sup>2+</sup> channels). Moreover, emodepside, but not its anthelmintically inactive enantiomer, decreased the response of the *H. contortus* LAT-1 receptor to latrotoxin [56].

#### PLOS PATHOGENS



Fig 2. Flow chart summarizing the most important achievements from the isolation of PF1022A to recent and future human trials. Milestones with pharmaceutical industry in lead are shown in blue, and milestones achieved with academia in lead are shown in yellow. BK, Big Potassium; DNDi, Drugs for Neglected Diseases initiative; LAT-1, latrophilin-like receptor.

https://doi.org/10.1371/journal.ppat.1009682.g002

In *C. elegans*, simultaneous deletion of both latrophilin paralogs, *lat-1* and *lat-2*, decreased the susceptibility of the pharynx to emodepside [57]. In contrast to *lat-1*, deletion of components of the signaling cascade downstream of the LAT-1 receptor, such as *egl-30* and *egl-8*, causes a decrease in emodepside susceptibility of both pharynx and body wall muscles, yet an *egl-30* gain-of-function mutant increases susceptibility [58]. The impact of defects downstream of LAT-1 on emodepside activity on the pharynx were always more severe than effects of LAT-1 [58]. This suggests that emodepside can directly exert its effects through LAT-1 on the pharynx, with upstream GPCRs augmenting emodepside's effects on body wall muscles.

Expression patterns of *lat-1* were determined in *C. elegans* using a fluorescent reporter gene construct [44,45,47]. In larvae, the promotor was active in neuronal and muscle cells around the pharynx. In contrast, promotor activity ceased in adult worms (which show the highest susceptibility to emodepside) and was limited to neuronal cells [23,59,60]. In contrast, Buxton and colleagues [61] showed expression of *lat-1* mRNA in muscle flaps of *A. suum* containing both muscle cells and axons of neuronal cells present in the dorsal and ventral cords. It currently remains unclear if these contrasting results are due to differences between nematode species or the different sensitivities of the methods used. It is also uncertain if emodepside is able to directly modulate LAT-1 from ascarids, since sequence identity between *A. suum* and *C. elegans* proteins is only 38% (54% similarity) [61].

In terms of understanding the mode of action of emodepside, it is important to note that *C. elegans* with double deletion of *lat-1* and *lat-2* are still nonviable in emodepside [58], suggesting that while latrophilin is a target of emodepside, other targets must also be key to its anthelmintic action. Knowledge regarding the different pathways involved in the effects of emodepside in nematode neurons and muscle cells is graphically summarized in Fig 3.

#### Voltage-gated and Ca<sup>2+</sup>-activated potassium channels (SLO-1)

In order to determine the molecular mediators of the latrophilin-independent effect of emodepside, a genetic screen in *C. elegans* revealed resistant worms with mutations in alleles of *slo-1* encoding a calcium-activated potassium channel [23,66]. SLO-1 channels are important for rapid repolarization of cells after depolarization during action potentials; a functional channel is made up of 4 subunits containing 7 transmembrane helices each (Fig 4). A functional null mutant of *slo-1* was found to develop and feed normally in the presence of a concentration of emodepside that impairs development, inhibits feeding, and paralyzes wild-type worms [67]. Interestingly *C. elegans* strains carrying *slo-1* gain-of-function mutations are sluggish, immotile, and retain eggs, phenocopying the effect of emodepside on *C. elegans* behavior [68]. These observations suggest that emodepside is a SLO-1 agonist, activating potassium currents to inhibit neuromuscular function. This has been confirmed through heterologous expression of *slo-1* in *C. elegans* pharyngeal muscle [63], HEK293 cells [60], and *Xenopus* oocytes [59,60].

The role of SLO-1 in the toxicity of emodepside to parasitic nematodes of companion animals and livestock was confirmed by the cloning of *slo-1* from *A. caninum* and *Cooperia oncophora* and its expression in the *C. elegans* functional null *slo-1* mutant, NM1968 *slo-1(js379)V*. Expression of *A. caninum* and *C. oncophora slo-1* genes in NM1968 was found to confer emodepside sensitivity to this otherwise unresponsive mutant and ameliorate the locomotor deficit of the *C. elegans slo-1* null background [62]. This provides strong evidence that emodepside exerts its anthelmintic action in both nematode species (*A. caninum* and *C. oncophora*) through the activation of SLO-1. The conservation of *slo-1* in the phylum Nematoda provides further evidence that this is the major determinant of its selective anthelmintic action (see Fig 5) [59].

To address the selective toxicity of emodepside in terms of its impact on the mammalian host, the actions of the closest human orthologue of *C. elegans slo-1, kcnma1*, was investigated. While emodepside was found to be a potent and prolonged activator of *C. elegans* SLO-1 heterologously expressed in HEK293 cells, it only caused transient activation of KCNMA1, followed by inhibition [60]. In the same study, emodepside also activated *Drosophila slo-1*, although its ability to modify the rectifying properties of the emodepside-induced current differed from that seen with *C. elegans slo-1* [60]. These functional differences in the interaction of emodepside with calcium- and voltage-activated potassium channels provide a mechanistic explanation for its selective anthelmintic action.



Fig 3. Graphical summary describing the signaling pathways potentially involved in Emo effects in nematodes. Signaling pathways are shown separately for pharynx (A) and body wall muscles (B). In several cases, there is no clear or unequivocal evidence for the presence/activity of the signaling component in the particular compartment. In these cases, question marks and dotted lines indicate these uncertainties. Emo directly interacts with the ionotropic GABA<sub>A</sub>-R UNC-49 [52], with LAT-1 [56] and with the voltage-gated Ca<sup>2+</sup>-activated potassium channel slowpoke big K<sup>+</sup> conductance channel (SLO-1) [59,60]. However, *Caenorhabditis elegans* lines with loss-of-function mutations in unc-49, lat-1, or both lat-1 and lat-2 genes display only modest decreases in their Emo sensitivity [55,57,58]. In contrast, loss of SLO-1 function in C. elegans leads to complete loss of Emo efficacy, which can be restored by expression of SLO-1 from parasitic nematodes, but only partially by the human orthologue [23,62,63]. This shows that activation of SLO-1 [59,60] is the major effect leading to Emo-induced paralysis as indicated by bold arrows. Activation of SLO-1 or GABAA-R UNC-49 can lead directly to membrane hyperpolarization, and, thus, to reduced excitation. UNC-49 is only expressed on body wall muscles (postsynaptic side) [64]. In contrast, SLO-1 channels can be found on presynaptic nerves innervating both pharynx and body wall muscles, as well as on body wall muscles themselves, being absent from pharyngeal muscles [63], LAT-1 (and potentially other, unidentified GPCRs [orphan receptors]) is assumed to exert its effects via a signal transduction cascade involving an hGp containing the EGL-30 Gqa subunit, leading to activation of the PLC EGL-8 [56-58]. PLC activity hydrolyses PIP2 to DAG and IP3. The latter leads to release of  $Ca^{2+}$  from intracellular stores. DAG and  $Ca^{2+}$  are both required to activate PKC- $\beta$ , which is known to be able to modulate SLO-1 responsiveness [61]. On the presynaptic side, DAG is further known to facilitate release of vesicles via the synaptobrevin SNB-1 and UNC-13, a protein essential for presynaptic vesicle release [57]. In C. elegans, these signaling pathways are also active in presynaptic neurons and postsynaptic body wall muscle cells [58]. In parasitic nematodes, the situation may be different since lat-1 mRNA was detected in Ascaris suum body wall flaps, suggesting that the receptor is expressed in motor neurons and/or body wall muscles [61]. Moreover, NO has been shown to increase potassium currents induced by Emo in A. suum muscle preparations [61]. It is currently unclear how NO exerts its effect since nematode soluble guanylate cyclases, in contrast to the mammalian enzymes, are not activated by NO [65]. DAG, diacyl glycerol; Emo, Emodepside; GABAA-R, γaminobutyric acid A receptor; GPCR, G protein-coupled receptor; hGp, heterotrimeric G protein; IP3, inositol 1,4,5-trisphosphate; LAT-1, latrophilin-1; NO, nitric oxide; PIP2, phosphatidylinositol 4,5-bisphosphate; PKC-β, protein kinase C β; PLC, phospholipase C; SLO-1, slowpoke big K<sup>+</sup> conductance channel.

https://doi.org/10.1371/journal.ppat.1009682.g003

A recent study used comparison of *C. elegans* wild strains with the N2 Bristol laboratory strain regarding emodepside responsiveness [72]. The authors found that there was substantial natural variation in susceptibility to emodepside. Some but not all of this variation could be explained by variation in the SLO-1 channel. The remaining variation was not explainable by variation in LAT-1, suggesting that there are additional genetic factors contributing to the level of emodepside susceptibility. These might include any of the signal transduction pathway components downstream of LAT-1 identified previously [58] and depicted in Fig.3 or additional, currently unidentified loci.

Taken together, the studies in *C. elegans* and parasitic nematodes have identified the calcium- and voltage-activated potassium channel SLO-1 as the major receptor mediating emodepside's anthelmintic action. The identity and properties of the pharmacophore for emodepside harbored by the channel remain a topic of speculation, although its physicochemical properties and profile of effects suggest a binding site within the membrane at a lipid:protein



Fig 4. Schematic drawings showing domain composition (A) and topology in the plasma membrane of a SLO-1 subunit (B). The scheme in (A) is drawn to scale based on *Onchocerca volvulus* SLO-1a (GenBank accession number: MW039265; the protein contains 7 transmembrane domains. The transmembrane domains S2–S4 contain charged amino acids and function as voltage sensor, while S5 and S6 are involved in building the pore. For this purpose, 4 subunits form a tetramer. The P loop is involved in tetramer formation and pore building. The 2 RCK domains and the Ca<sup>2+</sup> bowl are involved in regulation of conductance by intracellular Ca<sup>2+</sup> levels. P loop, pore-forming loop; PM, plasma membrane; RCK, regulator of potassium conductance; SLO-1, slowpoke big K<sup>+</sup> conductance channel.

https://doi.org/10.1371/journal.ppat.1009682.g004

interface [58]. However, elucidating the interaction of emodepside with SLO-1 was an important step toward understanding the molecular basis of its mode of action.

#### Emodepside as a filaricidal agent

There is an urgent need for an orally delivered macrofilaricidal drug suitable for MDA programs. In light of the success of emodepside as a veterinary anthelmintic [22], studies to evaluate the efficacy of this drug against filarial nematodes were planned. Initial studies employed in vivo filarial models in the southern multimammate mouse, Mastomys coucha [73]. Here, it was shown that a single dose of emodepside (100 mg/kg) applied as a spot-on formulation was able to substantially suppress microfilaremia in Acanthocheilonema viteae and Litomosoides sigmodontis infections for at least 100 days and in Brugia malayi infections for at least 150 days, when the drug was administered during the maturation of adult worms [73]. Only larval and preadult stages of A. viteae, and not those of L. sigmodontis and B. malayi, were affected by this treatment. A single dose of oral, subcutaneous, or spot-on treatment of 100 mg/kg emodepside has also been shown to eliminate all adult A. viteae in the same model [74]. In contrast, elimination of adult L. sigmodontis required repeated oral treatment of 100 mg/kg emodepside on 5 consecutive days, whereas adult B. malayi were not eliminated by this treatment schedule [74]. Nevertheless, severe pathological alterations in the intrauterine stages of all 3 parasite species were noted at a much lower single dose of emodepside, which is likely to explain the longterm effects of the drug on microfilarial density [73,74]. A direct microfilaricidal effect of emodepside was further shown in mice injected with microfilariae of Onchocerca lienalis at doses of  $>5 \times 1.56$  mg/kg [75].

The abovementioned in vivo studies showed that emodepside clears microfilariae of different filarial species, but has a less pronounced effect on the adult filarial burden. However, in vitro studies using a 4-day assay analyzing the motility of *L. sigmodontis* adult male and female worms showed complete inhibition of motility in both sexes at emodepside concentrations as low as  $1 \times 10^{-8}$  M. *L. sigmodontis* microfilariae and L3 larvae were less susceptible to emodepside, with complete motility inhibition demonstrated in a 4- and 3-day in vitro assay at an IC<sub>50</sub> of  $5 \times 10^{-9}$  M and  $3.5 \times 10^{-7}$ , respectively. Similarly, for microfilariae of *A. viteae* and *B*.



**Fig 5. Phylogenetic analysis of nematode SLO-1 channels. (A)** Maximum likelihood phylogram calculated from full-length Slo-1 proteins. Slo-1 sequences from the nematode species *Cel, Cbr, Cre, Pca, Hco, Con, Acan, Ovo, Ooc, Ogu, Bma, Dim, Tca, Peq, Asu, Tmu*, and *Tsu* were aligned with orthologs from *Dme, Aga, Phu, Dpu, Acal, Gal, Bta*, and *Hsa*, representing an outgroup, using MCoffee [69]. For *Onchocerca* spp. and *Brugia malayi*, all splice variants were included, but only the variants *slo-1a-c* for *Caenorhabditis elegans*. For all other species, only a single splice variant was used per gene. Accession numbers are available in <u>S1 Table</u>. ModelFinder [70] was applied to identify the optimal amino acid substitution model. The VT+G4 amino acid substitution model was used with a  $\Gamma$  shape alpha set to 0.4242 and the 4 substitution relative rate categories (with equal frequencies) of 0.0203, 0.1996, 0.7562, and 3.0239. Maximum likelihood trees were calculated with IQ-TREE [71] using 2,000 iterations of the Shimodaira–Hasegawa modification of the approximate likelihood ratio test (before the slash) and 2,000 ultrafast bootstrapping replicates (behind the slash). The scale bar represents 0.3 substitutions per site. (**B**) Enlarged cladogram showing the relationship of filarial SLO-1 proteins. For clarity, node support values

for this part of the tree are only indicated in B. Acal, Aplysia californica; Acan, Ancylostoma caninum; Aga, Anopheles gambiae; Asu, Ascaris suum; Bma, Brugia malayi; Bta, Bos taurus; Cbr, Caenorhabditis briggsae; Cel, Caenorhabditis elegans; Con, Cooperia oncophora; Cre, Caenorhabditis remanei; Dim, Dirofilaria immitis; Dme, Drosophila melanogaster; Dpu, Daphnia pulex; Gal, Gallus gallus; Hco, Haemonchus contortus; Hsa, Homo sapiens; Ogu, Onchocerca gutturosa; Ooc, Onchocerca ochengi; Ovo, Onchocerca volvulus; Pca, Pristionchus pacificus; Peq, Parascaris equorum; Phu, Pediculus humanus humanis; SLO-1, slowpoke big K<sup>+</sup> conductance channel; Tca, Toxocara canis; Tmu, Trichuris muris; Tsu, Trichuris suis.

https://doi.org/10.1371/journal.ppat.1009682.g005

*malayi*, complete inhibition of motility in a 4- and 3-day in vitro assay was demonstrated at an  $IC_{50}$  of  $1.5 \times 10^{-8}$  M and  $6.4 \times 10^{-8}$  M, respectively [24].

Further studies used several in vitro and in vivo systems that have been developed to study drug effects on *Onchocerca* spp. and *Brugia* spp. (see [5]). Utilizing these protocols, the efficacy of emodepside against 3 species of filarial parasites was evaluated, examining the effects of drug concentration and duration of exposure on adult male *Onchocerca gutturosa*, adult male and female *Brugia pahangi*, and microfilariae of *O. lienalis* in vitro [75]. The effects of drug dosage, formulation and route of administration on the survival of *O. lienalis* microfilariae in mice were also assessed [75].

O. gutturosa adult males obtained from the nuchal ligament connective tissues of naturally infected cattle were tested in a 5-day in vitro assay (concentration range  $2.95 \times 10^{-12}$  M to  $1.25 \times 10^{-5}$  M). Emodepside immobilized them at a concentration of  $\geq 4.8 \times 10^{-8}$  M, producing an EC<sub>50</sub> for motility of  $9 \times 10^{-10}$  M [35]. However, 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT)/formazan colorimetry results demonstrated that emodepside concentrations  $\leq 1.25 \times 10^{-5}$  M had no significant effect on macrofilarial viability, indicating that the worms were paralyzed, but not dead, after 5 days exposure to emodepside. A subsequent long-term in vitro assay confirmed that 5-day exposure to  $1.25 \times 10^{-5}$  M emodepside killed (measured both by motility and MTT colorimetry) *O. gutturosa* adult males after a total of 40 days of culture, indicating that worms did not recover following the initial 5-day exposure to emodepside [75].

Macrofilaricidal properties of emodepside were also evaluated against adult male and female *B. pahangi* in vitro [75]. These parasites were significantly less sensitive to emodepside than *O. gutturosa*, with *B. pahangi* parasites remaining active at a low level at  $1.25 \times 10^{-5}$  M in a 5-day assay, resulting in a sex-dependent motility EC<sub>50</sub> of  $6 \times 10^{-8}$  M for male worms and  $4.3 \times 10^{-7}$  M for females. Poor motility was also observed throughout a 10-day trial in which worms were transferred to a drug-free medium on day 5, with a similar partial effect seen on MTT values, indicating a significant drug effect, but not death [75].

Sex-dependent effects of emodepside have also recently been reported for *B. malayi* [76]. In particular, male and female worms have been shown to express distinctive splice variants of the emodepside target SLO-1, which differ greatly in their responsiveness to emodepside, presumably accounting for the major differences in emodepside susceptibility between the different sexes in this species.

A summary of studies investigating the activity of emodepside against filaria is provided in Table 1.

#### Development of emodepside for onchocerciasis

#### From animal health to neglected tropical diseases

Knowledge of the safety, efficacy, and pharmacokinetics (PK) of pharmaceuticals in animals can allow for a rapid transition into human use. Thus, veterinary anthelmintics should be considered as a priority for use against helminth parasites in humans, where there are still unmet needs [46,77].

Emodepside efficacy	Reference	
Adult worms: complete motility inhibition $(IC_{50} \ 1 \times 10^{-8} \ M)^a$ L3: complete motility inhibition $(IC_{50} \ 9 \times 10^{-9} \ M)^a$	[24]	
MF: complete motility inhibition $(IC_{50} 1 \times 10^{-9} M)^a$	[24]	
MF: complete motility inhibition $(IC_{50} \ 1 \times 10^{-9} \ M)^a$	[24]	
Adult male worms: complete motility inhibition $(EC_{50} 9 \times 10^{-10} \text{ M})^{b}$	[75]	
Adult male worms: reduced motility $(EC_{50} 6 \times 10^{-8} M)^{b}$	[75]	
Adult female worms: reduced motility (EC $_{50}$ 4.3 $\times$ $10^{-7}$ M) $^{\rm b}$		
MF: elimination for $\geq$ 100 days (100 mg/kg emodepside)	[73]	
Adult worms: elimination within 56 days (100 mg/kg emodepside)	[74]	
MF: elimination for $\geq$ 150 days (100 mg/kg emodepside)	[73]	
Adult worms: elimination within 56 days (5 $\times$ 100 mg/kg emodepside)	[74]	
Intrauterine stages: severe pathological alterations (25 mg/kg emodepside)	[74]	
	Image: Provide a series of the series of	Free regionReferenceAdult worms: complete motility inhibition $(IC_{50} 1 \times 10^{-8} M)^a$ [24]L3: complete motility inhibition $(IC_{50} 9 \times 10^{-9} M)^a$ [24]MF: complete motility inhibition $(IC_{50} 1 \times 10^{-9} M)^a$ [24]MF: complete motility inhibition $(IC_{50} 1 \times 10^{-9} M)^a$ [24]Adult male worms: complete motility inhibition $(EC_{50} 9 \times 10^{-10} M)^b$ [75]Adult male worms: reduced motility $(EC_{50} 6 \times 10^{-8} M)^b$ [75]Adult female worms: reduced motility $(EC_{50} 4.3 \times 10^{-7} M)^b$ [73]MF: elimination for $\geq 100$ days (100 mg/kg emodepside)[74]MF: elimination for $\geq 150$ days (100 mg/kg emodepside)[73]Adult worms: elimination within 56 days (5 $\times 100$ mg/kg emodepside)[74]Intrauterine stages: severe pathological alterations (25 mg/kg emodepside)[74]

#### Table 1. Efficacy studies for emodepside against filaria.

<sup>a</sup>4-day assay.

<sup>b</sup>4-day assay.

L3, third-stage larvae; MF, microfilariae.

https://doi.org/10.1371/journal.ppat.1009682.t001

Ivermectin is an example of applying the experience of anthelmintics developed for animal health to human NTDs. The drug was developed in the mid-1970s for control of livestock nematodes. By the late 1970s, it was observed that ivermectin was microfilaricidal against *Onchocerca* spp. in horses [78] and cattle [79] and prohibited the development of larval stages of the filarial nematode *Dirofilaria immitis* in dogs [80]. These observations led to the consideration of ivermectin for the control of *O. volvulus* [79]. The development of ivermectin for onchocerciasis, and, more recently for lymphatic filariasis, has been a highlight of efforts to control NTDs. For both filarial diseases, control programs rely on the microfilaricidal effect and temporal sterilization of female worms, which inhibit the release of new microfilariae for several months. For onchocerciasis and lymphatic filariasis control and elimination programs, ivermectin is donated under the Mectizan Donation Program established by Merck in 1987 [81].

There are a number of other examples of pharmaceuticals that were first developed for use in animal health ultimately having a huge impact on NTDs. Albendazole and mebendazole were first developed for animal health and have subsequently been used to control human soiltransmitted helminths (STHs) [82]. Hundreds of millions of doses per year of these anthelmintics are now donated for control of STH and lymphatic filariasis (albendazole) in humans. Another example, praziquantel, developed by Bayer and E. Merck (Merck Group), was patented initially as a veterinary drug. Subsequently, it was shown to be highly efficacious against human schistosomes, and it became essential in the fight against NTDs [83].

Bayer has a long history of discovering drugs for the treatment of NTDs. It developed a number of antiprotozoal drugs that are of considerable importance, including the suramin (Naganol) against African trypanosomiasis (discovered in 1924), the antimalarial drug chloroquine (in 1934 to 1937), and nifurtimox against Chagas disease (in 1972) [84]. Bayer-developed anthelmintics include antimosan against *Schistosoma mansoni* (in 1931), Miracil D (lucanthone) against some schistosomes (in 1940), and niclosamide (in 1953), which was used against tapeworms before the discovery of praziquantel [85]. Now, emodepside is the latest Bayer drug entering the field of tropical medicine. The DNDi, a collaborative, patient needs-driven, not-for-profit R&D organization, was founded in 2003. DNDi's primary focus is the development of drugs for the treatment of the most neglected diseases around the world [86]. In 2010, DNDi approved the inclusion of filarial diseases in its portfolio aiming to develop a safe, efficacious, affordable, and field-adapted drug targeting macrofilariae against onchocerciasis and/or lymphatic filariasis. One of the strategies pursued was to identify veterinary drugs and evaluate their suitability for development as macrofilaricides for human use [47].

Emodepside was originally identified as a potential macrofilaricide through a drug discovery effort aiming to identify macrofilaricidal drugs for filarial diseases sponsored by WHO Special Programme for Research and Training in Tropical Diseases (WHO-TDR) in the early 2000s [3,75,87]. In 2014, DNDi and Bayer signed an agreement for a collaboration on the development of emodepside as a potential treatment for onchocerciasis [47]. The scope of this agreement was to conduct a gap analysis of the veterinarian safety package as part of the completion of an investigational medicinal product dossier (IMPD). In addition, in vitro and in vivo studies were conducted with *L. sigmodontis* [24,88], a surrogate filarial nematode for efficacy against *O. volvulus* [89–91]. In the agreement, Bayer committed to conduct pharmaceutical development and supply of the drug, while DNDi was responsible for both nonclinical and clinical development of the molecule.

#### The Onchocerca ochengi/cattle model for preclinical proof of concept

*O. ochengi* is a sibling species to *O. volvulus* [92,93] infecting cattle, which is widely used for research into the basic biology of *Onchocerca* spp., drug effects, and vaccine candidates [94]. To assess the predictability of emodepside effects in the *O. ochengi* cattle model for human onchocerciasis, a recent study compared the emodepside susceptibility of SLO-1 of both *Onchocerca* species. A comparison of amino acid sequences from different isolated isoforms indicated that SLO-1 is highly conserved between both species (Fig 5). Moreover, when functionally expressed in the *Xenopus* oocyte expression system, *O. ochengi* and *O. volvulus* SLO-1 isoforms were all able to form homomeric channels, showing comparable levels of emodepside sensitivity [95].

In order to obtain reliable data using the O. ochengi/cattle model, it is crucial to achieve drug concentrations at the target site that are comparable with those in humans. The maximum tolerated dose (MTD) of emodepside in taurine Holstein cattle was found to be 1 mg/ kg, but 0.75 mg/kg was used conservatively in indicine (zebu) cattle in Cameroon to compensate for environmental stresses [96]. Analytical methods were also established to determine emodepside concentrations in skin and nodules containing adult O. ochengi after administration of different doses showing that an intravenous dose of 0.15 mg/kg in cattle was equivalent to 10 mg in humans, achieving the same local target concentration. Using this model, the density of microfilariae in the skin was not affected by a single 0.15 mg/kg dose of emodepside. In contrast, a single 0.75 mg/kg dose and 0.15 mg/kg given daily for 7 days transiently reduced microfilarial density [97]. In addition, 0.75 mg/kg emodepside given daily for 7 days resulted in an initial reduction, then a transient rise followed by complete clearance, of microfilariae in 4 out of 7 animals within 18 months of treatment. The motility of adult macrofilariae recovered from nodules was significantly affected over time at all emodepside doses for females, but only at the highest dose for males [97]. In 5 out of 7 cattle, including those with no microfilariae, the  $7 \times 0.75$  mg/kg treatment scheme caused death or sterility of female worms, demonstrating slow-acting macrofilaricidal and sterilizing effects [98].

#### Emodepside in Phase I human trials

Based on the available preclinical data, permission to proceed with Phase I trials was granted by the United Kingdom Medicines & Healthcare products Regulatory Agency (MHRA), and the study was initiated in December 2015 in the UK and completed in October 2018. Results of Phase I studies indicate that predicted efficacious levels could be reached safely in humans [99].

Phase I clinical trials with emodepside were comprised of single- (NCT02661178) and multiple-dose (NCT03383614) safety, tolerability, and PK studies and a relative bioavailability study (NCT03383523). In the single- (0.1 to 40 mg emodepside) and multiple-dose (5 or 10 mg) studies, an oral liquid formulation of emodepside was used, with the multidose escalation study also comparing emodepside 10 mg daily with 10 mg twice daily. In the bioavailability study, 2 immediate-release (IR) tablet emodepside formulations were compared with the liquid formulation (5 or 10 mg) [48]. As a liquid service formulation, emodepside was rapidly absorbed under fasting conditions, with dose-proportional increases in plasma concentrations at doses from 1 mg to 40 mg. The half-life during the first 24 hours after dosing was around 11 hours, followed by a terminal elimination half-life >500 hours. Emodepside was less bioavailable in the fed state. Emodepside was well tolerated overall, with no major safety concerns. The rate of absorption was slower, and the peak serum concentration (C<sub>max</sub>) was slightly lower with the amorphous solid dispersion tablets than with the liquid service formulation (NCT03383614, NCT02661178, and NCT03383523). These data enabled us to select a field-adapted tablet formulation that will open the way for further clinical development of emodepside in individuals with onchocerciasis. In terms of macrofilaricidal activity, a target concentration corresponding to the in vitro minimum inhibitory concentration against L. sigmodontis of 100 ng/ml was assumed to be required. Simulation studies were used to model dosing regimens that would achieve such a target site concentration. A 15-mg dose with the gastrosoluble tablet is predicted to provide exposure that will achieve the target concentration for clinical efficacy [48,99].

A Phase II clinical trial will take place in Hohoe, Ghana. The study is a randomized, double-blind, parallel group trial to investigate emodepside in subjects with *O. volvulus* infection, comprising 2 parts. In part 1, the safety, tolerability, pharmacodynamics, PK, and dose–response relationship for efficacy (proof of concept) was investigated. In part 2, the efficacy of selected doses, safety, tolerability, and PK will be investigated [100].

# Beyond onchocerciasis—Should emodepside be considered for human soil-transmitted helminths?

STH infections, including *Ascaris lumbricoides*, *Trichuris trichiura*, and the 2 hookworm species *Ancylostoma duodenale* and *Necator americanus*, are highly prevalent and responsible for a considerable public health burden [101]. Chemotherapy is the strategy of choice for treatment and control of these infections. However, there is a need to discover and develop alternative drugs since treatment options are limited, drug resistance is a concern, and, importantly, the 2 most widely used drugs, the benzimidazoles albendazole and mebendazole, have limited activity against *T. trichiura* when used in a single-dose regimen [102]. Given the limited drug discovery pipeline for STH infections, veterinary anthelmintics are an attractive starting point for crossover development for the treatment of these diseases, particularly given the fact that almost all drugs against these pathogens have been introduced initially into the veterinary market (see above).

Emodepside could be an excellent drug candidate for the treatment of STH infections. In S2 Table, studies conducted to date investigating the activity of emodepside against *Trichuris* 

spp., ascarids, *Strongyloides ratti*, and hookworm species have been summarized. Briefly, emodepside is highly active against *T. vulpis* in dogs at single oral doses of 0.5 to 2 mg/kg [37]. In the *T. muris* mouse model, an ED<sub>95</sub> of 24.5 mg/kg for oral treatment was calculated [103], while a recent study reported an ED<sub>50</sub> of 1.2 mg/kg and complete clearance of the parasites at 75 mg/kg for a single oral treatment [104]; for comparison, an ED<sub>50</sub> of 4.7 mg/kg was recently determined for a single dose of oxantel pamoate against *T. muris* in mice [105]. A single dose of emodepside was shown to be highly effective against ascarids in cats (*Toxocara cati* and *Toxascaris leonina*) and dogs (*Toxocara canis* and *T. leonina*) [34,38]. Remarkably, it is also able to suppress galactogenic transmission of *T. cati* [106]. Studies with emodepside against hookworms were conducted with the mouse and rat roundworms *Heligmosomoides bakeri* and *Nippostrongylus brasiliensis* [107], the zoonotic parasites *Ancylostoma tubaeforme* [37] and *A. ceylanicum* [104] in cats and *A. caninum* in dogs [39,108], as well as the human parasite *N. americanus* in a hamster model [104]. Importantly, emodepside also revealed activity against *Strongyloides*, a neglected helminth species [107].

In summary, based on the preclinical and clinical studies in animals conducted over the past 20 years, it can be anticipated that emodepside would be a useful addition to the small drug armamentarium for human STH infections. As described below, Phase II clinical trials are planned with emodepside against *T. trichiura* and hookworm infections. In parallel, confirmatory preclinical studies should take place with the single product against *Strongyloides stercoralis* in dogs and against *Ascaris* spp. in pigs, as well as for migrating larvae in the mouse model.

#### Current status of development program and outlook

Using a multiple-dose therapeutic scheme, emodepside will be evaluated for its macrofilaricidal and long-term sterilizing activity in onchocerciasis patients. Emodepside's activity against *T. trichiura* and hookworms will also be evaluated in parallel.

Although much work remains to be done, current progress suggests that emodepside could be a valuable future weapon in the arsenal to fight against important NTDs caused by parasitic nematodes.

#### Supporting information

**S1 Fig. Geographical distribution and status of PC in endemic countries in 2017.** PC, preventive chemotherapy.

(JPG)

S2 Fig. Life cycle of Onchocerca volvulus. Blackflies of the genus Simulium transmit L3 of O. volvulus onto human skin from where larvae then actively penetrate into the bite wound (1). In subcutaneous tissues, larvae develop into adult filariae (2). Adults reside in nodules in subcutaneous connective tissues (3) and can live there for up to 15 years. Nodules can contain multiple relatively shorter male (19–42 mm) and longer female (33–50 cm) worms. Female worms are viviparous and can produce microfilariae for about 9 years. Microfilariae are unsheathed and have a length of 220–360  $\mu$ m and a diameter of 5–9  $\mu$ m. They can survive up to 2 years in the human host. Although microfilariae can be found occasionally in peripheral blood, urine, and sputum, they are typically found in the skin and in the lymphatics of connective tissues (4). Transmission to the intermediate host occurs when a blackfly ingests microfilariae during blood feeding (5). Microfilariae penetrate the wall of the blackfly's midgut and migrate through the hemocoel to the thoracic muscles (6) where the development from first-stage (7) into L2 and L3 (8) occurs. The iL3 migrate to the blackfly's proboscis (9) and can

infect another human when the vector takes a blood meal (1). iL3, third-stage infective larvae; L2, second-stage larvae; L3, third-stage larvae.

(JPG)

**S1 Table. SLO-1 protein database entries included in the phylogenetic analysis.** SLO-1, slowpoke big K<sup>+</sup> conductance channel. (DOCX)

S2 Table. Efficacy studies for emodepside against *Trichuris* spp., ascarids, hookworms, and *Strongyloides ratti*.

(DOCX)

#### References

- 1. World Health Organization. Onchocerciasis. Fact Sheet. 2019 Jun 14 [cited 2021 Feb 12]. https://www.who.int/news-room/fact-sheets/detail/onchocerciasis.
- WHO/TDR Scientific Working Group. RNA interference as a means of identifying drug targets for Filariasis. [cited 2021 Feb 14]. https://www.who.int/tdr/publications/disease/lymphatic-filariasis/en/.
- 3. Kuesel AC. Research for new drugs for elimination of onchocerciasis in Africa. Int J Parasitol Drugs Drug Resist. 2016; 6:272–86. https://doi.org/10.1016/j.ijpddr.2016.04.002 PMID: 27693536
- Mathers CD, Ezzati M, Lopez AD. Measuring the burden of neglected tropical diseases: the global burden of disease framework. PLoS Negl Trop Dis. 2007; 1:e114. <u>https://doi.org/10.1371/journal.pntd.</u> 0000114 PMID: 18060077
- Townson S, Ramirez B, Fakorede F, Mouries M-A, Nwaka S. Challenges in drug discovery for novel antifilarials. Expert Opin Drug Discovery. 2007; 2:S63–73. <u>https://doi.org/10.1517/17460441.2.S1.S63</u> PMID: 23489034
- WHO. Progress report on the elimination of human onchocerciasis, 2017–2018. Wkly Epidemiol Rec. 2018; 47:633–48.
- Plaisier AP, van GJ, Remme J, Habbema JD. The reproductive lifespan of *Onchocerca volvulus* in West African savanna. Acta Trop. 1991; 48:271–84. https://doi.org/10.1016/0001-706x(91)90015-c PMID: <u>1674401</u>
- Hedtke SM, Kuesel AC, Crawford KE, Graves PM, Boussinesq M, Lau CL, et al. Genomic Epidemiology in Filarial Nematodes: Transforming the Basis for Elimination Program Decisions. Front Genet. 2020; 10:1282. https://doi.org/10.3389/fgene.2019.01282 PMID: 31998356
- Basáñez M-G, Pion SDS, Churcher TS, Breitling LP, Little MP, Boussinesq M. River blindness: a success story under threat? PLoS Med. 2006; 3:e371. <u>https://doi.org/10.1371/journal.pmed.0030371</u> PMID: 17002504
- Taylor MJ, Hoerauf A, Bockarie M. Lymphatic filariasis and onchocerciasis. Lancet. 2010; 376:1175– 85. https://doi.org/10.1016/S0140-6736(10)60586-7 PMID: 20739055
- Little MP, Breitling LP, Basáñez M-G, Alley ES, Boatin BA. Association between microfilarial load and excess mortality in onchocerciasis: an epidemiological study. Lancet. 2004; 363:1514–21. https://doi. org/10.1016/S0140-6736(04)16151-5 PMID: 15135599
- Geary TG, Mackenzie CD, Silber SA. Flubendazole as a macrofilaricide: History and background. PLoS Negl Trop Dis. 2019; 13:e0006436. https://doi.org/10.1371/journal.pntd.0006436 PMID: 30650160
- Prichard RK, Geary TG. Perspectives on the utility of moxidectin for the control of parasitic nematodes in the face of developing anthelmintic resistance. Int J Parasitol Drugs Drug Resist. 2019; 10:69–83. https://doi.org/10.1016/j.ijpddr.2019.06.002 PMID: 31229910
- Walker M, Specht S, Churcher TS, Hoerauf A, Taylor MJ, Basáñez M-G. Therapeutic efficacy and macrofilaricidal activity of doxycycline for the treatment of river blindness. Clin Infect Dis. 2015; 60:1199–207. https://doi.org/10.1093/cid/ciu1152 PMID: 25537873
- Lustigman S, Makepeace BL, Klei TR, Babayan SA, Hotez P, Abraham D, et al. Onchocerca volvulus: The road from basic biology to a vaccine. Trends Parasitol. 2018; 34:64–79. https://doi.org/10.1016/j. pt.2017.08.011 PMID: 28958602
- Boussinesq M, Gardon J, Gardon-Wendel N, Kamgno J, Ngoumou P, Chippaux JP. Three probable cases of *Loa loa* encephalopathy following ivermectin treatment for onchocerciasis. Am J Trop Med Hyg. 1998; 58:461–9. https://doi.org/10.4269/ajtmh.1998.58.461 PMID: 9574793.

- Chippaux JP, Boussinesq M, Gardon J, Gardon-Wendel N, Ernould JC. Severe adverse reaction risks during mass treatment with ivermectin in loiasis-endemic areas. Parasitol Today. 1996; 12:448–50. https://doi.org/10.1016/0169-4758(96)40006-0 PMID: 15275280.
- Bah GS, Ward EL, Srivastava A, Trees AJ, Tanya VN, Makepeace BL. Efficacy of three-week oxytetracycline or rifampin monotherapy compared with a combination regimen against the filarial nematode Onchocerca ochengi. Antimicrob Agents Chemother. 2014; 58:801–10. <u>https://doi.org/10.1128/AAC.</u> 01995-13 PMID: 24247133
- 19. Abong RA, Amambo GN, Chounna Ndongmo PW, Njouendou AJ, Manuel R, Beng AA, et al. Differential susceptibility of *Onchocerca volvulus* microfilaria to ivermectin in two areas of contrasting history of mass drug administration in Cameroon: relevance of microscopy and molecular techniques for the monitoring of skin microfilarial repopulation within six months of direct observed treatment. BMC Infect Dis. 2020; 20:726. https://doi.org/10.1186/s12879-020-05444-2 PMID: 33008333
- G-FINDER/Policy Cures Research. Global funding of innovation for neglected diseases. 2019 [cited 2021 Feb 12]. https://s3-ap-southeast-2.amazonaws.com/policy-cures-website-assets/app/uploads/ 2020/02/07161934/GF-6pSummary2019.pdf.
- Harder A, Holden-Dye L, Walker R, Wunderlich F. Mechanisms of action of emodepside. Parasitol Res. 2005; 97(Suppl. 1):S1–10. https://doi.org/10.1007/s00436-005-1438-z PMID: 16228263
- Krücken J, Harder A, Jeschke P, Holden-Dye L, O'Connor V, Welz C, et al. Anthelmintic cyclooctadepsipeptides: complex in structure and mode of action. Trends Parasitol. 2012; 28:385–94. https://doi. org/10.1016/j.pt.2012.06.005 PMID: 22858281
- Guest M, Bull K, Walker RJ, Amliwala K, O'Connor V, Harder A, et al. The calcium-activated potassium channel, SLO-1, is required for the action of the novel cyclo-octadepsipeptide anthelmintic, emodepside, in *Caenorhabditis elegans*. Int J Parasitol. 2007; 37:1577–88. <u>https://doi.org/10.1016/j.ijpara.</u> 2007.05.006 PMID: 17583712
- Kulke D, Townson S, Bloemker D, Frohberger S, Specht S, Scandale I, et al. Comparison of the in vitro susceptibility to emodepside of microfilariae, third stage larvae and adult worms of related filarial nematodes. Am J Trop Med Hyg. 2017; 97:563. https://doi.org/10.4269/ajtmh.16-0968 PMID: 28829737
- Sasaki T, Takagi M, Yaguchi T, Miyadoh S, Okada T, Koyama M. A new anthelmintic cyclodepsipeptide, PF1022A. J Antibiot. 1992; 45:692–697. <u>https://doi.org/10.7164/antibiotics.45.692</u> PMID: 1624372
- Kachi S, Ishih A, Terada M. Effects of PF1022A on adult Angiostrongylus cantonensis in the pulmonary arteries and larvae migrating into the central nervous system of rats. Parasitol Res. 1995; 81:631–7. https://doi.org/10.1007/BF00931838 PMID: 8570576
- von Samson-Himmelstjerna G, Harder A, Schnieder T, Kalbe J, Mencke N. In vivo activities of the new anthelmintic depsipeptide PF 1022A. Parasitol Res. 2000; 86:194–9. <u>https://doi.org/10.1007/</u> s004360050031 PMID: 10726989
- 28. Wang M, Watanabe N, Shomura T, Ohtomo H. Effects of PF1022A on *Nippostrongylus brasiliensis* in rats and *Hymenolepis nana* in mice. Jpn J Parasitol. 1995; 440:306–10.
- Kachi S, Terada M, Hashimoto H. Effects of amorphous and polymorphs of PF1022A, a new antinematode drug, on Angiostrongylus costaricensis in mice. Jpn J Pharmacol. 1998; 77:235–45. https:// doi.org/10.1254/jjp.77.235 PMID: 9717771
- Nwosu U, Vargas M, Harder A, Keiser J. Efficacy of the cyclooctadepsipeptide PF1022A against *Heligmosomoides bakeri* in vitro and in vivo. Parasitology. 2011; 138:1193–201. <u>https://doi.org/10.1017/S003118201100076X PMID: 21756418</u>
- Wang M, Watanabe N, Shomura T, Ohtomo H. Effects of PF1022A from *Mycelia sterilia* on *Trichinella spiralis*. Jpn J Parasitol. 1994; 43:346–50.
- Conder GA, Johnson SS, Nowakowski DS, Blake TE, Dutton FE, Nelson SJ, et al. Anthelmintic profile of the cyclodepsipeptide PF1022A in in vitro and in vivo models. J Antibiot (Tokyo). 1995; 48:820–3. https://doi.org/10.7164/antibiotics.48.820 PMID: 7592027
- Altreuther G, Borgsteede FHM, Buch J, Charles SD, Cruthers L, Epe C, et al. Efficacy of a topically administered combination of emodepside and praziquantel against mature and immature *Ancylostoma tubaeforme* in domestic cats. Parasitol Res. 2005; 97(Suppl. 1):S51–7. <u>https://doi.org/10.1007/</u> s00436-005-1444-1 PMID: 16228275
- Altreuther G, Schimmel A, Schroeder I, Bach T, Charles S, Kok DJ, et al. Efficacy of emodepside plus praziquantel tablets (Profender tablets for dogs) against mature and immature infections with *Toxocara canis* and *Toxascaris leonina* in dogs. Parasitol Res. 2009; 105(Suppl.):S1–8. https://doi.org/10. 1007/s00436-009-1489-7 PMID: 19575219
- Nicolay F, Harder A. von Samson-Himmelstjerna G, Mehlhorn H. Synergistic action of a cyclic depsipeptide and piperazine on nematodes. Parasitol Res. 2000; 86:982–92. <u>https://doi.org/10.1007/</u> pl00008530 PMID: 11133114

- Mehlhorn H, Schmahl G, Frese M, Mevissen I, Harder A, Krieger K. Effects of a combinations of emodepside and praziquantel on parasites of reptiles and rodents. Parasitol Res. 2005; 97(Suppl. 1):S65– 9. https://doi.org/10.1007/s00436-005-1446-z PMID: 16228277
- Petry G, Altreuther G, Wolken S, Swart P, Kok DJ. Efficacy of emodepside plus toltrazuril oral suspension for dogs (Procox, Bayer) against *Trichuris vulpis* in naturally infected dogs. Parasitol Res. 2013; 112(Suppl.):133–8. https://doi.org/10.1007/s00436-013-3287-5 PMID: 23756961
- Reinemeyer CR, Charles SD, Buch J, Settje T, Altreuther G, Cruthers L, et al. Evaluation of the efficacy of emodepside plus praziquantel topical solution against ascarid infections (*Toxocara cati* or *Toxascaris leonina*) in cats. Parasitol Res. 2005; 97(Suppl. 1):S41–50. <u>https://doi.org/10.1007/s00436-005-1443-2 PMID: 16228274</u>
- 39. Schimmel A, Altreuther G, Schroeder I, Charles S, Cruthers L, Ketzis J, et al. Efficacy of emodepside plus praziquantel tablets (Profender tablets for dogs) against mature and immature adult *Ancylostoma caninum* and *Uncinaria stenocephala* infections in dogs. Parasitol Res. 2009; 105(Suppl.):S9–16. https://doi.org/10.1007/s00436-009-1490-1 PMID: 19575220
- Schmahl G, Mehlhorn H, Harder A, Klimpel S, Krieger KJ. Efficacy of a combination of emodepside plus praziquantel against larval and adult stages of nematodes (*Trichuris muris, Angiostrongylus cantonensis*) in rodents. Parasitol Res. 2007; 101:77–84. <u>https://doi.org/10.1007/s00436-006-0445-z</u> PMID: 17235546
- Traversa D, Milillo P, Di A, Lohr B, Iorio R, Pampurini F, et al. Efficacy and safety of emodepside 2.1%/ praziquantel 8.6% spot-on formulation in the treatment of feline aelurostrongylosis. Parasitol Res. 2009; 105(Suppl.):S83–9. https://doi.org/10.1007/s00436-009-1499-5 PMID: 19575229
- Gaens D, Leithäuser C, Hamann M, Geyer J. Adverse drug reactions after administration of emodepside/praziquantel (Profender) in an MDR1-mutant Australian shepherd dog: Case report. Front Vet Sci. 2019; 6:296. https://doi.org/10.3389/fvets.2019.00296 PMID: 31555677
- 43. Jimenez Castro PD, Mansour A, Charles S, Hostetler J, Settje T, Kulke D, et al. Efficacy evaluation of anthelmintic products against an infection with the canine hookworm (*Ancylostoma caninum*) isolate Worthy 4.1F3P in dogs. Int J Parasitol Drugs Drug Resist. 2020; 13:22–7. https://doi.org/10.1016/j. ijpddr.2020.04.003 PMID: 32403053
- von G, Harder A, Sangster NC, Coles GC. Efficacy of two cyclooctadepsipeptides, PF1022A and emodepside, against anthelmintic-resistant nematodes in sheep and cattle. Parasitology. 2005; 130:343– 7. https://doi.org/10.1017/s0031182004006523 PMID: 15796017
- 45. Harder A, Schmitt-Wrede H-P, Krücken J, Marinovski P, Wunderlich F, Willson J, et al. Cyclooctadepsipeptides—an anthelmintically active class of compounds exhibiting a novel mode of action. Int J Antimicrob Agents. 2003; 22:318–31. https://doi.org/10.1016/s0924-8579(03)00219-x PMID: 13678839
- 46. Geary TG, Woo K, McCarthy JS, Mackenzie CD, Horton J, Prichard RK, et al. Unresolved issues in anthelmintic pharmacology for helminthiases of humans. Int J Parasitol. 2010; 40:1–13. <u>https://doi.org/ 10.1016/j.ijpara.2009.11.001 PMID: 19932111</u>
- **47.** Drugs for Neglected Diseases Initiative. Filaria: river blindness. Emodepside. 2020 Feb 29 [cited 2021 Feb 12]. https://dndi.org/research-development/portfolio/emodepside.
- 48. Gillon J-Y, Dennison J, van den Berg F, Delhomme S, Dequatre Cheeseman K, Pêna Rossi C, et al. Safety, tolerability and pharmacokinetics of emodepside, a potential novel treatment for onchocerciasis (river blindness), in healthy male subjects. Brit Jnl Clinical Pharma. 2021; 1–12. https://doi.org/10. 1111/bcp.14816
- Sasaki T, Takagi M, Yaguchi T, Miyadoh S, Okada T, Koyama M. A new anthelmintic cyclodepsipeptide, PF1022A. J Antibiot (Tokyo). 1992; 45:692–7. https://doi.org/10.7164/antibiotics.45.692 PMID: 1624372
- Ohyama M, Iinuma K, Isogai A, Suzuki A. Total synthesis of the anthelmintic cyclodepsipeptide, PF1022A. Biosci Biotechnol Biochem. 1994; 58:1193–4. https://doi.org/10.1271/bbb.58.1193
- Pleiss U, Harder A, Turberg A, Londershausen M, linuma K, Mencke N, et al. Synthesis of a radiolabeled cyclodepsipeptide [3H-methyl]PF1022A. J Label Compd Radiopharm. 1996; 38:61–9. <a href="https://doi.org/10.1002/(SICI)1099-1344(199601)38:1>61::AID-JLCR814>3.0.CO;2-8">https://doi.org/10.1002/(SICI)1099-1344(199601)38:1>61::AID-JLCR814>3.0.CO;2-8</a>
- 52. Chen W, Terada M, Cheng JT. Characterization of subtypes of gamma-aminobutyric acid receptors in an Ascaris muscle preparation by binding assay and binding of PF1022A, a new anthelmintic, on the receptors. Parasitol Res. 1996; 82:97–101. https://doi.org/10.1007/s004360050077 PMID: 8825201
- Martin RJ. Electrophysiological effects of piperazine and diethylcarbamazine on Ascaris suum somatic muscle. Br J Pharmacol. 1982; 77:255–65. https://doi.org/10.1111/j.1476-5381.1982.tb09294.x PMID: 7139188
- Martin RJ, Harder A, Londershausen M, Jeschke P. Anthelmintic actions of the cyclic depsipeptide PF1022A and its electrophysiological effects on muscle cells of *Ascaris suum*. Pest Sci. 1996; 48:343– 9.

- 55. Miltsch SM, Krücken J, Demeler J, Janssen IJI, Krüger N, Harder A, et al. Decreased emodepside sensitivity in unc-49 γ-aminobutyric acid (GABA)-receptor-deficient *Caenorhabditis elegans*. Int J Parasitol. 2012; 42:761–70. https://doi.org/10.1016/j.ijpara.2012.05.009 PMID: 22727682
- Saeger B, Schmitt-Wrede HP, Dehnhardt M, Benten WP, Krücken J, Harder A, et al. Latrophilin-like receptor from the parasitic nematode *Haemonchus contortus* as target for the anthelmintic depsipeptide PF1022A. FASEB J. 2001; 15:1332–4. https://doi.org/10.1096/fj.00-0664fje PMID: 11344131
- Willson J, Amliwala K, Davis A, Cook A, Cuttle MF, Kriek N, et al. Latrotoxin receptor signaling engages the UNC-13-dependent vesicle-priming pathway in *C. elegans*. Curr Biol. 2004; 14:1374–9. https://doi.org/10.1016/j.cub.2004.07.056 PMID: 15296755
- Holden-Dye L, Crisford A, Welz C, von G, Walker RJ, O'Connor V. Worms take to the slo lane: a perspective on the mode of action of emodepside. Invert Neurosci. 2012; 12:29–36. https://doi.org/10. 1007/s10158-012-0133-x PMID: 22539031
- 59. Kulke D, von G, Miltsch SM, Wolstenholme AJ, Jex AR, Gasser RB, et al. Characterization of the Ca2 +-gated and voltage-dependent K+-channel Slo-1 of nematodes and its interaction with emodepside. PLoS Negl Trop Dis. 2014; 8:e3401. https://doi.org/10.1371/journal.pntd.0003401 PMID: 25521608
- 60. Crisford A, Ebbinghaus-Kintscher U, Schoenhense E, Harder A, Raming K, O'Kelly I, et al. The cyclooctadepsipeptide anthelmintic emodepside differentially modulates nematode, insect and human calcium-activated potassium (SLO) channel alpha subunits. PLoS Negl Trop Dis. 2015; 9:e0004062. https://doi.org/10.1371/journal.pntd.0004062 PMID: 26437177
- Buxton SK, Neveu C, Charvet CL, Robertson AP, Martin RJ. On the mode of action of emodepside: slow effects on membrane potential and voltage-activated currents in *Ascaris suum*. Br J Pharmacol. 2011; 164:453–70. https://doi.org/10.1111/j.1476-5381.2011.01428.x PMID: 21486286
- Welz C, Krüger N, Schniederjans M, Miltsch SM, Krücken J, Guest M, et al. SLO-1-channels of parasitic nematodes reconstitute locomotor behaviour and emodepside sensitivity in *Caenorhabditis elegans slo-1* loss of function mutants. PLoS Pathog. 2011; 7:e1001330. https://doi.org/10.1371/journal. ppat.1001330 PMID: 21490955
- Crisford A, Murray C, O'Connor V, Edwards RJ, Kruger N, Welz C, et al. Selective toxicity of the anthelmintic emodepside revealed by heterologous expression of human KCNMA1 in *Caenorhabditis elegans*. Mol Pharmacol. 2011; 79:1031–43. https://doi.org/10.1124/mol.111.071043 PMID: 21415309
- Bamber BA, Beg AA, Twyman RE, Jorgensen EM. The Caenorhabditis elegans unc-49 locus encodes multiple subunits of a heteromultimeric GABA receptor. J Neurosci. 1999; 19:5348–59. https://doi.org/ 10.1523/JNEUROSCI.19-13-05348.1999 PMID: 10377345
- Morton DB, Hudson ML, Waters E, O'Shea M. Soluble guanylyl cyclases in *Caenorhabditis elegans*: NO is not the answer. Curr Biol. 1999; 9:R546–7. <u>https://doi.org/10.1016/s0960-9822(99)80349-2</u> PMID: 10469574
- Amliwala K. Molecular and genetic determinants of the inhibitory action of emodepside on *C. elegans* muscle. PhD Thesis. The University of Southampton; 2005.
- Bull K, Cook A, Hopper NA, Harder A, Holden-Dye L, Walker RJ. Effects of the novel anthelmintic emodepside on the locomotion, egg-laying behaviour and development of *Caenorhabditis elegans*. Int J Parasitol. 2007; 37:627–36. https://doi.org/10.1016/j.ijpara.2006.10.013 PMID: 17157854
- Holden-Dye L, O'Connor V, Hopper NA, Walker RJ, Harder A, Bull K, et al. SLO, SLO, quick, quick, slow: calcium-activated potassium channels as regulators of *Caenorhabditis elegans* behaviour and targets for anthelmintics. Invert Neurosci. 2007; 7:199–208. https://doi.org/10.1007/s10158-007-0057-z PMID: 17962986
- Notredame C, Higgins DG, Heringa J. T-Coffee: A novel method for fast and accurate multiple sequence alignment. J Mol Biol. 2000; 302:205–17. https://doi.org/10.1006/jmbi.2000.4042 PMID: 10964570
- Kalyaanamoorthy S, Minh BQ, Wong TKF, von A, Jermiin LS. ModelFinder: fast model selection for accurate phylogenetic estimates. Nat Methods. 2017; 14:587–9. <u>https://doi.org/10.1038/nmeth.4285</u> PMID: 28481363
- Nguyen L-T, Schmidt HA, von A, Minh BQ. IQ-TREE: a fast and effective stochastic algorithm for estimating maximum-likelihood phylogenies. Mol Biol Evol. 2015; 32:268–74. <u>https://doi.org/10.1093/molbev/msu300</u> PMID: 25371430
- 72. Wit J, Rodriguez BC, Andersen EC. Natural variation in *Caenorhabditis elegans* responses to the anthelmintic emodepside. Int J Parasitol Drugs Drug Resist. 2021; 16:1–8. <u>https://doi.org/10.1016/j.ijpddr.2021.04.001</u> PMID: 33878514
- 73. Zahner H, Taubert A, Harder A, von Samson-Himmelstjerna G. Effects of Bay 44–4400, a new cyclodepsipeptide, on developing stages of filariae (*Acanthocheilonema viteae, Brugia malayi, Litomosoides sigmodontis*) in the rodent *Mastomys coucha*. Acta Trop. 2001; 80:19–28. https://doi.org/10. 1016/s0001-706x(01)00144-9 PMID: 11495640

- 74. Zahner H, Taubert A, Harder A. von Samson-Himmelstjerna G. Filaricidal efficacy of anthelmintically active cyclodepsipeptides. Int J Parasitol. 2001; 31:1515–22. https://doi.org/10.1016/s0020-7519(01) 00263-6 PMID: 11595239
- 75. Townson S, Freeman A, Harris A, Harder A. Activity of the cyclooctadepsipeptide emodepside against Onchocerca gutturosa, Onchocerca lienalis and Brugia pahangi. Am J Trop Med Hyg. 2005; 73(6; Suppl. 1): abstract 28076.
- 76. Kashyap SS, Verma S, Voronin D, Lustigman S, Kulke D, Robertson AP, et al. Emodepside has sexdependent immobilizing effects on adult *Brugia malayi* due to a differentially spliced binding pocket in the RCK1 region of the SLO-1 K channel. PLoS Pathog. 2019; 15:e1008041. https://doi.org/10.1371/ journal.ppat.1008041 PMID: 31553770
- 77. Geary TG, Woods DJ, Williams T, Nwaka S. Target Identification and Mechanism-Based Screening for Anthelmintics: Application of Veterinary Antiparasitic Research Programs to Search for New Antiparasitic Drugs for Human Indications. In: Antiparasitic and Antibacterial Drug Discovery: From Molecular Targets to Drug Candidates. Selzer PM, editor. Weinheim, Germany: Wiley; 2009.
- 78. Klei TR, Torbert BJ, Ochoa R. Efficacy of ivermectin (22,23-dihydroavermectin B1) against adult Setaria equine and microfilariae of Onchocerca cervicalis in ponies. J Parasitol. 1980; 66:859–61. PMID: 6894003
- Campbell WC. History of avermectin and ivermectin, with notes on the history of other macrocyclic lactone antiparasitic agents. Curr Pharm Biotechnol. 2012; 13:853–65. <u>https://doi.org/10.2174/138920112800399095</u> PMID: 22039784
- Jackson RF, Seymour WG. In: Otto GF, editor. Proceedings of the Heartworm Symposium, 1980. Edwardsville, KS: Veterinary Medicine Publishing Co.; 1981. p. 131–6.
- 81. Mectizan Donation Program. Overview. 2021 [cited 2021 Feb 12]. https://mectizan.org/what/overview.
- Olliaro P, Seiler J, Kuesel A, Horton J, Clark JN, Don R, et al. Potential drug development candidates for human soil-transmitted helminthiases. PLoS Negl Trop Dis. 2011; 5:e1138. <u>https://doi.org/10.1371/journal.pntd.0001138</u> PMID: 21695247
- 83. Lee SH, Chai JY. Praziquantel in the treatment of trematode and cestode infections. J Korean Soc Chemother. 1985; 3:95–118.
- Haberkorn A, Harder A, Greif G. Milestones of protozoan research at Bayer. Parasitol Res. 2001; 87:1060–2. https://doi.org/10.1007/s004360100406 PMID: 11763440
- Harder A. Milestones of helmintic research at Bayer. Parasitol Res. 2002; 88:477–80. https://doi.org/ 10.1007/s00436-001-0551-x PMID: 12049469
- Drugs for Neglected Diseases initiative. About us. 2020 [cited 2021 Feb 12]. <u>https://dndi.org/about/who-we-are.</u>
- WHO/TDR. Tropical Disease Research: Progress 2003–2004. 2021 [cited 2021 Feb 12]. https://www. who.int/tdr/publications/about-tdr/progress-reports/progress-report-03-04/en.
- Specht S, Mackenzie C, Townson S, Don R, Hoerauf A. Drug discovery and development for the treatment and control of filariasis: repurposing emodepside. Am J Trop Med Hyg. 2014; 91(5; Suppl. 1): abstract 1209.
- Allen JE, Adjei O, Bain O, Hoerauf A, Hoffmann WH, Makepeace BL, et al. Of mice, cattle, and humans: The immunology and treatment of river blindness. PLoS Negl Trop Dis. 2008; 2:e217. <a href="https://doi.org/10.1371/journal.pntd.0000217">https://doi.org/10.1371/journal.pntd.0000217</a> PMID: 18446236
- 90. Hübner MP, Martin C, Specht S, Koschel M, Dubben B, Frohberger SJ, et al. Oxfendazole mediates macrofilaricidal efficacy against the filarial nematode *Litomosoides sigmodontis* in vivo and inhibits *Onchocerca* spec. motility in vitro. PLoS Negl Trop Dis. 2020; 14:e0008427. <u>https://doi.org/10.1371/journal.pntd.0008427</u> PMID: 32628671
- Hübner MP, Koschel M, Struever D, Nikolov V, Frohberger SJ, Ehrens A, et al. vivo kinetics of *Wolbachia* depletion by ABBV-4083 in *L. sigmodontis* adult worms and microfilariae. PLoS Negl Trop Dis. 2019; 13:e0007636. https://doi.org/10.1371/journal.pntd.0007636 PMID: 31381563
- Lefoulon E, Giannelli A, Makepeace BL, Mutafchiev Y, Townson S, Uni S, et al. Whence river blindness? The domestication of mammals and host-parasite co-evolution in the nematode genus Onchocerca. Int J Parasitol. 2017; 47:457–70. https://doi.org/10.1016/j.ijpara.2016.12.009 PMID: 28344097
- Cotton JA, Bennuru S, Grote A, Harsha B, Tracey A, Beech R, et al. The genome of Onchocerca volvulus, agent of river blindness. Nat Microbiol. 2016; 2:16216. https://doi.org/10.1038/nmicrobiol.2016. 216 PMID: 27869790
- 94. Makepeace BL, Tanya VN. 25 Years of the Onchocerca ochengi Model. Trends Parasitol. 2016; 32:966–78. https://doi.org/10.1016/j.pt.2016.08.013 PMID: 27665524

- Hahnel SR, Rufener L, Heisler I, Schneckener S, Makepeace B, Glenschek-Sieberth MKD. Heterologously expressed SLO-1 isoforms from *Onchocerca volvulus* and *Onchocerca ochengi* exhibit similar levels of emodepside sensitivity. Am J Trop Med Hyg. 2020; 103(Suppl. 5):378.
- 96. Fieseler H, Krebber R, Schneckener S, Fraatz K, Bah GS, Townson S, et al. Emodepside as macrofilaricide for the treatment of river blindness: An interdisciplinary and collaborative approach towards a meaningful proof of concept for the preclinical package. Am J Trop Med Hyg. 2020; 103(Suppl. 5):105.
- Schneckener S, Makepeace BL, Kulke D, Glenschek-Sieberth M. Model based characterization of the pharmacodynamic effect of emodepside on *Onchocerca ochengi* in African cattle. Am J Trop Med Hyg. 2020; 103(Suppl. 5):106.
- Bah GS, Schneckener S, Hahnel SR, Bayang NH, Fieseler H, Schmuck GM, et al. Emodepside targets SLO-1 channels of *Onchocerca ochengi* and induces broad anthelmintic effects in a bovine model of onchocerciasis. PLoS Pathog. 2021; 17(6). https://doi.org/10.1371/journal.ppat.1009601
- Assmus F, Hoglund RM, Scandale I, Tarning J. Drug development for the treatment and control of onchocerciasis: a population pharmacokinetic analysis of emodepside (BAY 44–4400) in healthy volunteers. Am Soc Trop Med Hyg. 2019; 101(Suppl. 5):391–2.
- Drugs for Neglected Diseases Initiative. R&D Portfolio Update February 2019: DNDi Filarial diseases programme. 2019 Feb 21 [cited 2021 Feb 12]. https://dndi.org/news/2019/filaria-rnd-status-2019/.
- 101. Pullan RL, Brooker SJ. The global limits and population at risk of soil-transmitted helminth infections in 2010. Parasit Vectors. 2012; 5(81). https://doi.org/10.1186/1756-3305-5-81 PMID: 22537799
- 102. Keiser J, Utzinger J. The drugs we have and the drugs we need against major helminth infections. Adv Parasitol. 2010; 73:197–230. https://doi.org/10.1016/S0065-308X(10)73008-6 PMID: 20627144
- 103. Kulke D, Krücken J, Harder A, von G. Efficacy of cyclooctadepsipeptides and aminophenylamidines against larval, immature and mature adult stages of a parasitologically characterized trichurosis model in mice. PLoS Negl Trop Dis. 2014; 8:e2698. <u>https://doi.org/10.1371/journal.pntd.0002698</u> PMID: 24587460
- 104. Karpstein T, Pasche V, Häberli C, Scandale I, Neodo A, Keiser J. Evaluation of emodepside in laboratory models of human intestinal nematode and schistosome infections. Parasit Vectors. 2019; 12:226. https://doi.org/10.1186/s13071-019-3476-x PMID: 31088525
- 105. Keiser J, Tritten L, Silbereisen A, Speich B, Adelfio R, Vargas M. Activity of oxantel pamoate monotherapy and combination chemotherapy against *Trichuris muris* and hookworms: revival of an old drug. PLoS Negl Trop Dis. 2013; 7:e2119. https://doi.org/10.1371/journal.pntd.0002119 PMID: 23556013
- 106. Wolken S, Schaper R, Mencke N, Kraemer F, Schnieder T. Treatment and prevention of vertical transmission of *Toxocara cati* in cats with an emodepside/praziquantel spot-on formulation. Parasitol Res. 2009; 105(Suppl.):S75–81. https://doi.org/10.1007/s00436-009-1498-6 PMID: 19575228
- 107. Harder A, Samson-Himmelstjerna G. Activity of the cyclic depsipeptide emodepside (BAY 44–4400) against larval and adult stages of nematodes in rodents and the influence on worm survival. Parasitol Res. 2001; 87:924–8. https://doi.org/10.1007/s004360100479 PMID: 11728017
- 108. Schimmel A, Schroeder I, Altreuther G, Settje T, Charles S, Wolken S, et al. Efficacy of emodepside plus toltrazuril (Procox() oral suspension for dogs) against *Toxocara canis*, *Uncinaria stenocephala* and *Ancylostoma caninum* in dogs. Parasitol Res. 2011; 109(Suppl.):S1–8. https://doi.org/10.1007/ s00436-011-2397-1 PMID: 21739370